Author's response to reviews

Title: Gender difference in suicidal expressions and its determinants among young people in Cambodia, a post-conflict country.

Authors:

Jegannathan Bhoomikumar (drbhoomikumari@gmail.com)
Gunnar Kullgren (gunnar.kullgren@psychiat.umu.se)

Version: 5 Date: 15 March 2011

Author's response to reviews:

According to suggestions:
We have added Both authors have read and approved the final manuscript
Introduction has been changes to Background
A few other typographical changes are made